Author’s response to reviews

Title: Sexual and Gender Minorities Rights in Latin America and the Caribbean: A multi-country evaluation

Authors:

Monica Malta (monica.malta@camh.ca)
Reynaldo Cardoso (reynaldoluiu@gmail.com)
Luiz Montenegro (luiz.montenegro@gmail.com)
Jaqueline Gomes de Jesus (jaqueline.jesus@ifrj.edu.br)
Michele Seixas (m.seixaso@gmail.com)
Bruna Benevides (bruna-marx@hotmail.com)
Maria das Dores Silva (ninamds@bol.com.br)
Sara LeGrand (sara.legrand@duke.edu)
Kathryn Whetten (kathwhetten@gmail.com)

Version: 3 Date: 15 Oct 2019

Author’s response to reviews:

Dear Editors,

We are re-submitting our manuscript entitled “Sexual and Gender Minorities Rights in Latin America and the Caribbean: A multi-country evaluation.” (IHHR-D-18-00118R1). The revised manuscript addresses all essential revisions suggested by the editor:

1) Competing interests: As Monica Malta is Associate Editor for BMC Public Health, please declare this in the Competing interests section of the Declarations.

We revised this section, including the participation in editorial boards of both BMC Public Health and BMC International Health and Human Rights (Monica Malta).

2) Authors’ contributions – missing authors MDDS and KW

We revised this section, describing the individual contribution of all authors.

3) Clean manuscript
A final manuscript without any changes/comments was included.

We thank the editor and reviewers for their careful evaluation and contribution to our work.